BodyWire-HCI: Enabling New Interaction Modalities by Communicating Strictly During Touch Using Electro-Quasistatic Human Body Communication

SHOVAN MAITY, DAVID YANG, SCOTT STANTON REDFORD, DEBAYAN DAS, BAIBHAB CHATTERJEE, and SHREYAS SEN, Purdue University

Communication during touch provides a seamless and natural way of interaction between humans and ambient intelligence. Current techniques that couple wireless transmission with touch detection suffer from the problem of selectivity and security, i.e., they cannot ensure communication only through direct touch and not through close proximity. We present *BodyWire-HCI*, which utilizes the human body as a wire-like communication channel, to enable human–computer interaction, that for the first time, demonstrates selective and physically secure communication strictly during touch. The signal leakage out of the body is minimized by utilizing a novel, low frequency Electro-QuasiStatic Human Body Communication (EQS-HBC) technique that enables interaction strictly when there is a conductive communication path between the transmitter and receiver through the human body. Design techniques such as capacitive termination and voltage mode operation are used to minimize the human body channel loss to operate at low frequencies and enable EQS-HBC. The demonstrations highlight the impact of *BodyWire-HCI* in enabling new human–machine interaction modalities for variety of application scenarios such as secure authentication (e.g., opening a door and pairing a smart device) and information exchange (e.g., payment, image, medical data, and personal profile transfer) through touch (https://www.youtube.com/watch?v=Uwrig2XQIH8).

CCS Concepts: • Hardware \rightarrow Wireless devices; Wireless integrated network sensors; Emerging interfaces; • Human-centered computing \rightarrow Interaction paradigms; Interaction devices;

Additional Key Words and Phrases: Human Body Communication (HBC), Human Computer Interaction (HCI), HBC-HCI, Security, Strictly Touch Based Communication, Electro Quasi-Static HBC

ACM Reference format:

Shovan Maity, David Yang, Scott Stanton Redford, Debayan Das, Baibhab Chatterjee, and Shreyas Sen. 2020. *BodyWire-HCI*: Enabling New Interaction Modalities by Communicating Strictly During Touch Using Electro-Quasistatic Human Body Communication. *ACM Trans. Comput.-Hum. Interact.* 27, 6, Article 39 (November 2020), 25 pages.

https://doi.org/10.1145/3406238

© 2020 Association for Computing Machinery.

1073-0516/2020/11-ART39 \$15.00

https://doi.org/10.1145/3406238

ACM Transactions on Computer-Human Interaction, Vol. 27, No. 6, Article 39. Publication date: November 2020.

This work was supported in part by the National Science Foundation CRII Award under Grant CNS 1657455, National Science Foundation Career Award under grant CCSS 1944602, and in part by the Air Force Office of Scientific Research YIP Award under Grant FA9550-17-1-0450.

Authors' addresses: S. Maity, D. Yang, S. S. Redford, D. Das, B. Chatterjee, and S. Sen, Purdue University, 465 Northwestern Avenue, West Lafayette, Indiana, 47907.

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than ACM must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

1 INTRODUCTION

Decades of rapid advancement in semiconductor technology has enabled computing in cheap, small form factor everyday devices. Human users are now submerged in a sea of computers, with which they are constantly interacting. User interaction with the surrounding environment is one of the key aspects of ubiquitous computing. Communication between devices during a touch-based interaction will significantly enhance the effect of a touch event and open up new interaction modalities in human–computer interaction (HCI). The standard method of achieving this is through coupling touch (Figure 2(a)) and communication (Figure 2(b)) separately. That can be done by utilizing a touch sensor (example: capacitive touch sensor) along with some wireless communication protocol such as Bluetooth, Near Field Communication (NFC) (Figure 2(c)). These protocols use radio waves, utilizing air as the communication medium. Wireless radio wave communication suffers from the problem of security and selectivity, as the signal gets transmitted through air medium and is available to any device within a certain range. As a result, information may get communicated when the devices are in close proximity, even before touch. Thus, communication strictly during a touch event i a secure and selective manner is difficult to achieve with radio communication.

Imagine if it is possible to communicate securely and selectively, strictly during a touch event (Figure 2(d)). Human Body Communication (HBC) [56, 65, 70] potentially provides such a secure, selective, and natural way of touch-based interaction between users and the environment, turning everyday objects into an interactive computing platform. HBC uses the human body as the communication channel for interactions between devices on and around the body. The transmitter couples the signal into the body through a metal electrode. The signal then goes through the skin layer and gets transmitted through the conductive tissues and fluids within the body and is picked up at the receiver end when it is in contact with the skin. As a communication method, HBC provides the following three key advantages over radio wave communication: (a) security, (b) selectivity, and (c) enhanced device lifetime.

The security aspect of HBC comes from the fact that the signal transmission is contained primarily within or in close proximity to the human body. Hence, an attacker cannot snoop any ongoing HBC transmission without either touching or coming in close proximity to the person. In radio wave transmission, this is not possible, as the signal is radiated from the transmitter into the air medium and any malicious attacker who is within the range of the transmission can pick up the transmitted signals. The human body as a communication medium can potentially provide this additional physical layer of security which is not present in radio wave communication.

The selectivity aspect of HBC also follows from the signal confinement within the body of the person, who is transmitting. Hence, multiple receivers in close proximity will not be affected by each other, as the signal intended for a particular receiver will not get picked up by an adjacent receiver. However, for radio wave wireless communication, selectivity between adjacent receivers is strongly dependent on the transmit power and sensitivity of the receiver. Communications intended for one receiver will be effectively broadcasted to other receivers in close proximity. In HCI scenarios, which require high selectivity between multiple spatially close sensors. Wireless radio communication can trigger multiple receiver devices at one time. Appropriate modality of HBC can provide appropriate selectivity in these scenarios, solving the problem of multiple receivers triggering simultaneously.

Radio wave communication suffers high loss around the body due to absorption and multi-path reflection effects. Moreover, the operating frequency of radio wave communication needs to be in the range of 100s of MHz to GHz due to the physical size limitation imposed of the antennas. Small devices are restricted to having small antennas, which makes the operating frequency higher.



Fig. 1. *BodyWire-HCI* utilizes Electro-Quasistatic HBC technique to communicate through the body without radiating signal out of the body enabling selective and secure communication strictly through touch. (a) A wearable transmitter communicating with a receiver connected to a laptop which enables human computer interaction. (b) Demonstration showing selectivity of *BodyWire-HCI* transmission. (c) Information exchange between a wearable and a computer. (d) Secure Authentication through *BodyWire-HCI*. Link to demo: https://www.youtube.com/watch?v=Uwrig2XQIH8.

The high operating frequency increases the power consumption in radio wave communication reducing the battery life of small form factor wearable devices. In HBC operation, no antenna is required for signal transmission, as the signal is coupled to the body. Hence, the operating frequency of HBC is not limited by any physical parameter. Therefore, systems utilizing HBC can operate at lower frequencies which make them power efficient compared to wireless systems using radio waves, subsequently increasing the lifetime of the devices and systems.

These potential advantages, coupled with the natural and seamless integration of touch in everyday life, make HBC a promising alternative to radio wave communication for acting as the communication medium during interactions between a wearable and any computing device in the surrounding environment (Figure 1). Coupling communication strictly with a touch event can open up new interaction modalities utilizing everyday conductive objects as potential touch sensors. HCI through HBC can enable different applications such as authenticating a door with touch through transfer of a key, transferring data between a computer and smart watch, and information exchange between smart watches of two people during a meeting or some other social gathering [56]. In this article, we develop an HBC system that utilizes electro-quasistatic transmission of electrical signals through the human body to achieve enhanced selectivity and security by minimizing signal leakage out of the body into the air medium. The receiver and transmitter communicate with each other only when both devices are in direct contact with the human body and there is no communication even when the devices are in very close proximity to the body. Electro-quasistatic Human Body Communication (EQS-HBC) transmission requires operating at a low frequency range to minimize signal leakage out of the human body. Previous studies characterizing the human body channel as a communication medium have shown the body to be a high loss channel at low frequencies, making it undesirable to operate HBC systems at low frequencies.



Fig. 2. Different possible interaction scenarios between a body worn wearable and off-body computing device. (a) Only Touch: a touch sensitive surface is used to detect a touch event. (b) Only communication: a wearable and machine communicating with each other through wireless communication. (c) Communication while touch through wireless communication radiates the signal, resulting in communication even without touch and also enabling data snooping from an attacker. (d) Communication during touch through HBC confining the signal within the body enabling communication strictly during touch.

Hence, certain system design techniques have to be followed to enable HBC signal transmission at low loss at these low frequencies to minimize signal leakage. The main contributions of this article are the following:

- A. Technical Advancement of EQS-HBC for HCI:
 - -EQS-HBC HCI Device Development: This article develops the first EQS-HBC device for HCI applications, using the fundamental theory developed previously [14]. The channel loss introduced by the body determines the amount of signal at the receiver end and is dependent on different factors specific to the device design, such as the form factor, the size of the ground plane, the proximity between the device ground and signal plane, location of the body, and quality of signal coupling into the body. We design a wearable EQS-HBC device taking these different factors into account.
 - —EQS-HBC Channel Characterization for reliable communication: Detailed characterization of the channel is carried out for different conditions such as transmitter/receiver electrode size (Figure 7) and posture of the human body (Figure 11). These are necessary to determine the requirements of the receiver such as sensitivity and reliable data rate of communication. This helps design a receiver that can work under different conditions. This article has focused on these evaluations to further the understanding of the EQS-HBC channel and make informed design decisions for the *BodyWire-HCI* prototype, such that it is possible to reliably send data strictly through touch.
 - —Interaction Aware Signal Leakage Evaluation: The selectivity property requires further evaluation of signal leakage on multiple electrodes in close proximity to each other. This article also focuses on how the user's technique of interacting with the devices can potentially affect these signal leakages and hence the selectivity property.
- B. Computer Human Interaction Demonstration through EQS-HBC:
 - EQS-HBC Demonstration: This article develops an EQS-HBC HCI device to show the first Commercial off-the-shelf (COTS) component-based demonstration of EQS-HBC. The different design considerations for such a system is also discussed in detail. The

designed EQS-HBC system is used to demonstrate three applications towards computer-human interaction.

- Strictly Touch Based, Selective Communication: This article demonstrates how the signal confinement property of EQS-HBC can be extended to strictly touch–based communication and selectivity. These properties are then applied to demonstrate three applications towards computer human interaction scenarios.
- Simultaneous multi-user multi-touch identification: The low signal leakage of *BodyWire-HCI* enables simultaneous multi-user multi-touch user identification, not possible with current HBC technology because of cross-device interference due to leakage.

The rest of the article is organized as follows: Section 2 provides background information about how the human body can be utilized as a communication channel for interactions strictly through touch and discusses the previous demonstrations of HBC identifying the open problems. Section 3 discusses the fundamental techniques used in *BodyWire-HCI* system to enable HBC transmission with minimal leakage. Section 4 characterizes the human body channel as a communication medium and determines the range of loss with variation among subjects and parameters such as electrode size. Section 5 provides the design goals of the prototype, while Section 6 describes the safety aspects and limitations of the *BodyWire-HCI*. The demonstrations of the *BodyWire-HCI* system are shown in Section 7 and Section 8 concludes the article.

2 MOTIVATION

2.1 Touch-Based Communication

Touch has been widely used as a medium for humans to communicate with devices around them. Touchscreen displays on mobile devices and computers provide an interactive way to communicate with those devices. Touch has also been used as an interaction technique for 3D interactive environments [5, 22] and haptics. Touch-based interaction has even been recommended for older users due to its limited cognitive, spatial and attentional demand [8, 67]. However, adding interaction along with touch, which enhances the effect of touch, is seen as a promising application area [4, 9]. One example of such application is using mobile phones to interact with larger public displays [23]. Paying through mobile applications is also one such example where the user has to authenticate himself/herself by touching a fingerprint sensor and subsequently use NFC-based communication to finish the payment. Enhancing touch-based interaction by combining it with communication can increase the convenience of such applications significantly. In this article, we utilize HBC to enable communication strictly during touch, enhancing the effect of touch-based interactions.

BodyWire-HCI essentially utilizes electrical signal transmission through the body to provide a way for humans to interact with devices. There is a large body of work in the HCI field which utilizes the interaction of human body with electrical signals or electric fields. Finding Common Ground [20] offers a comprehensive summary of Capacitive Touch Sensing in HCI. Field mice [60], Smartskin [53], Diamond Touch [16], Swiss-cheese extended [19], Opencapsense [34], DGTS [21], Platypus [66], Tile Track [62], Cohn et al. [12], and Gong et al. [18], and the like use electric field sensing to identify the location of single/multiple touches using an array of transmitter and receiver devices. *BodyWire-HCI* doesn't enable us to detect touch location and it is not possible to get the resolution achieved through Electric Field Sensing (EFS). Instead, it augments touch sensing by adding communication during touch. Implementations such as Skintrack [68], Aurasense [69], High5 [32], Living Wall [6], iSphere [35], Multi-Touch Skin [50], Humantenna [14], and Your noise is my command [13] enable gesture recognition utilizing the interaction between an electric field and the human body. Biometric authentication/identification has also been implemented



Fig. 3. Signal Attenuation characteristics of HBC. The transmitter sends out a 3.3V signal, which is then transmitted through the body and gets attenuated to a 30mV signal at the receiver end, showing a human body channel loss of about 40dB. The channel loss is dependent on the size of the device, location of the device on the body and body posture and can go up to 70dB.

through touch-based interaction between humans and computers, such as Fiberio [26], Biometric Authentication [29], On Demand Biometrics [27], Wearable key [46], Bodyprint [28], Capacitive Fingerprinting [24], Touche [72], Handsdown [55], and Carpicio [33]. These implementations use biometrics for identifying an individual. We demonstrate authentication/ identification as a possible application of *BodyWire-HCI*. However, in this case a password/key is sent through the covert channel through the body and allows an additional method of authentication and augment with biometrics to achieve multi-factor authentication. TAP [51], Touchcomm [63, 64], Earthlings Attack! [61], Wearable key [46], Skintrack [68], Aurasense [69], CarpetLAN [17], and so on, utilize the body as the communication medium to interact between devices and is the closest and most directly related set of work compared to *BodyWire-HCI*. However, no other HBC work has shown communication strictly during touch because of significant radiation during in body data transmission through HBC. *BodyWire-HCI* is the first system to achieve strictly touch–based communication through its low frequency operation and progresses state of the art HBC implementations. In the next subsection we look into the fundamentals of HBC and discuss a few of the circuit and system level implementations of previous HBC systems.

2.2 Background – HBC Basics

HBC, also commonly referred to as Body Coupled Communication or Intra-Body Communication was first proposed by Zimmerman [70] as a method of communication for devices in Personal Area Networks. A battery-operated transmitter is used to couple a small amount of modulated displacement current into the human body and transmit it through to the receiver. This kind of HBC technique is termed as Capacitive HBC, since the closed loop circuit in this scenario is formed by a capacitive return path through coupling between the transmitter and receiver ground planes with the surrounding environment. However, this results in the signal getting attenuated at the receiver end, as shown in Figure 3. Wegmueller et al. [65] proposed Galvanic HBC, where the transmitted signal is applied to the human body between two electrodes connected to the body and picked up at the receiver end through two similar electrodes. However, it has been established that in Galvanic HBC the signal gets attenuated more for longer distance communication within the body (example: left arm to right arm/arm to torso). Recently there have been studies about Magnetic HBC [52], where magnetic fields are used to communicate between devices on the body. However, this requires bulky coils at the transmitter and receiver end to pick up the magnetic field. Hence, the receiver and transmitter device sizes become big and are not suitable for the design of a wearable form factor device. Hence for our demonstration, we have used Capacitive HBC as the type of HBC transmission.

Electric Field Sensing	Gesture Recognition	Biometrics	Human Body Communication (HBC)	
[12], [16], [18], [19], [20], [21], [34], [53], [60], [62]	[6], [13], [14], [32], [35], [50], [68], [69]	[24], [26], [27], [28], [29], [33], [46], [55], [72]	HCI: [17], [25], [30], [46], [49], [51], [61], [63], [64], [68], [69] Circuit: [3], [10], [11], [37], [40], [41], [54] Channel Char: [1], [39], [42], [44]	Electro- QuasiStatic HBC (Strictly Touch)
				This Work

Table 1. Categorization of Previous Literature Focused on Touch-based Communication

BodyWireHCI is the first system which enables Strictly touch–based communication through signal confinement. Comparison of *BodyWire-HCI* with State-of-the-Art Touch–Based Communication Techniques.

2.3 Related Work – HBC

There have been previous studies showing both system level and circuit level implementations of HBC based systems for interactions between wearable devices and computers as classified in Table 1. Microchip Technology's Bodycom system [73] shows interactions between a wearable transmitter on a pocket and other everyday objects such as a car/home door, power drill, and the like. The devices are able to communicate when the transmitter is kept in the pocket, showing that direct touch is not a strict requirement for communication. [74] shows the demonstration of a ring being used as an intermediate device for exchanging data between two computers. The data is downloaded from a computer into the ring while touching a pair of electrodes attached to the computer. But in this scenario the communication is done by the touch of two fingers, which forms a closed path between the transmitter and receiver through the electrodes, making it unusable for wearable devices, like a smartwatch/smart-band. The demonstrations in [30] show the transfer of an image between a mobile transmitter and a receiver terminal connected to the wall. It also shows that even when the transmitter and receiver are a few centimeters away from each other, they can still communicate with just the transmitter being held in hand and the receiver not in contact with the body. [71] shows applications where the color of the clothing of a person or a ball changes depending on the color displayed at the transmitter end. Various demonstrations of cabinet unlocking and authentication on doors are shown in [49]. However, in these cases, the demonstrations work when the person is not in direct contact with both the transmitter and receiver devices. [63] shows multiple applications utilizing HBC for HCI, such as playing interactive games by touching or walking on custom electronic floor tiles. The authors show that even when the person wearing the transmitter is standing on one floor tile, there is signal leakage into the adjacent floor tiles. In addition to this the experiments show communication between the wearable and the floor tiles through the shoes of the user, proving that touch is not a strict necessity for communication. The demonstrations in [25] show the usage of commodity devices such as smartphone and touchpads to transmit secret keys through the body for user authentication. The achievable data rate in this scenario is a few hundred bits per second, which is sufficient for the authentication demonstrations shown but not sufficient for information transfer.

The studies discussed so far focus on system level demonstration of HBC. There have also been previous studies [3, 10, 11, 37, 38, 43, 54], which focus on the design of custom HBC integrated circuits. Most of these designs also operate at high frequencies, making them prone to signal leakage out of the body.



Fig. 4. (a) EQS-HBC enables signal confinement within body. So, there is no signal leakage out of the body. (b) This results in communication strictly during touch.

2.4 Open Problems

The previous demonstrations of HBC show information exchange when a path exists between the transmitter and the receiver. However most of them suffer from signal leakage, which results in the receiver triggering even when there is no direct physical contact with the receiver. [30] shows information getting transferred even when the transmitter is a few centimeters away from the receiver. The authors in [63] show that signal leaks into the adjacent floor tiles even when the person with the transmitter is standing on a separate floor tile. Other demonstrations also show communication when the device is not in direct contact with the body. This shows that there is significantly high signal leakage out of the body, resulting in communication between the devices even without direct touch. As a result, these demonstrations do not have the property of information exchange strictly during touch.

However, the goal of this article is to design systems such that a touch event is the necessary and sufficient condition for signal transmission (Figure 4). This can also be represented as

Touch
$$\rightarrow$$
 Communication (1)

$$!Touch \rightarrow !Communication$$
 (2)

Each of the individual equations provides a necessary condition for communication. When combined together, these equations provide the necessary and sufficient condition for strictly touchbased communication. However, it does not differentiate between the touch of an intended device or a malicious attacker. An attacker can snoop the signal only if he can make contact with the person. Strictly touch-based communication reduces the possibility of signal leakage to an attacker significantly compared to other scenarios, where the signal is available even when the attacker is



Fig. 5. Measurement of signal leakage into the air, as the receiver is moved away from the body. (a) Measurement setup showing leakage measurement and on-body signal transmission measurement. (b) Plot of measured voltage with distance away from body. (c) Correlation between the measured and transmitted voltage over distance at different angles. The correlation is <0.5 even for distances a few cm away from the body, highlighting the difficulty for an attacker to snoop an ongoing transmission [14].

away from the person. This can provide a security benefit compared to previous implementations of HBC.

3 FUNDAMENTAL TECHNIQUES

We design the *BodyWire-HCI* prototype following certain system level techniques and parameter choices to minimize signal leakage out of the body and ensure selectivity and security during HCIs. The key system level design techniques utilized to achieve this are: (a) EQS-HBC, (b) Capacitive receiver termination, and (c) Voltage mode transmission and reception. In this section, we discuss the effect of each of these techniques in achieving the goal of higher selectivity and security.

3.1 Electro-Quasistatic (EQS) HBC

In EQS-HBC, the signal transmission between the transmitter and the receiver occurs through electric fields, contained within the human body [15]. The signal propagation occurs primarily through the low impedance tissue layers underneath the skin. Propagation of electro-magnetic waves is not the primary mode of communication in this scenario. Hence, there is minimal signal leakage out of the body during ongoing signal transmission within the body. The low frequency operation of the HBC system enables electro-quasistatic operation and is the primary reason for the confinement of the transmitted signals within the body. Figure 5 shows the measured voltage at varying distances from the body when there is a signal transmission within the body at 1 MHz frequency. It can be seen that the amount of signal leakage in the air falls below 10 millivolts even 1 centimeter away from the human body, showing signal confinement primarily within the body. The signal received only a few centimeters away from the body also shows minimal correlation with the transmitted signal. Hence, it will be nearly impossible for an attacker to successfully decipher the ongoing data transmission even from a distance of a few centimeters away from the body. However, to achieve electro-quasistatic operation, it is important to operate at low frequencies to minimize signal radiation. Previous studies characterizing the human body [1, 2, 7, 31, 36, 39, 44, 47, 52] had shown high signal attenuation at low frequencies, making it an undesirable frequency band for HBC transmission. However, for our current system design, we use capacitive termination and voltage mode operation to reduce the human body channel loss at low frequencies and extend the bandwidth of the human body. This enables the use of the body as a wire-like communication channel, even in low frequencies and opens up the possibility of using it as a secure, selective communication channel.

This article develops a HCI device using EQS-HBC for a wearable form factor. This article further investigates the EQS-HBC channel under different conditions. The signal confinement property of EQS-HBC is evaluated through signal leakage measurements on multiple electrodes in close proximity to each other. The understanding developed from the channel characteristics and signal leakage measurements are utilized to show the first COTS component-based demonstration of EQS-HBC. Moreover, the signal confinement property of EQS-HBC is extended to selective, strictly touch-based communication to demonstrate applications towards computer human interaction. The selectivity property of *BodyWire-HCI* enables modalities which require *simultaneous multi-user multi-touch identification*, not possible through existing HBC or wireless communication techniques.

3.2 Capacitive Termination

The value and type of termination impedance of the human body channel at the receiver end is an important parameter in determining the operable frequency range of the HBC system. Resistive termination is the commonly used termination methodology in most previous studies. However, in our design, the termination at the receiver end is done through a high impedance capacitor. Because of the capacitive return path between the transmitter and the receiver, using a capacitive termination at the receiver end creates a loss response independent of frequency [38, 40, 41, 57–59]. This reduces low frequency loss and enables HBC at these frequencies. In our current demonstration, specific design techniques are followed such that the receiver's input impedance and hence the termination impedance seen by the channel is primarily capacitive. Also, the receiver is designed to minimize the input capacitance, which reduces the channel loss.

3.3 Voltage Mode Transmission and Reception

Previous HBC system designs primarily use signal power as the metric for communication between devices. However, power transmission requires low impedance termination at the receiver end, and therefore is not suitable for low frequency signal transmission through HBC. In our demonstrations we use voltage mode operation, where the transmitter sends out voltage through the system and the receiver is designed to maximize the received voltage. Voltage mode operation requires high impedance termination at the receiver end, enhancing the amount of signal received for low frequency HBC operation [41, 57]. Hence, design of voltage mode systems also help reduce the low frequency loss. This in turn enables EQS-HBC, providing signal confinement within the body.

4 HUMAN BODY COMMUNICATION CHARACTERIZATION

In order to reliably realize the goals of selective and secure EQS-HBC for a wide range of applications, *BodyWire-HCI* was designed after careful user experimentation to ensure system functionality independent of human factors and also provide security and selectivity. This requires characterizing the human body as a communication channel along with the signal leakage aspect of EQS-HBC. Channel characterization is done through an experiment on channel loss variation among different users. In addition, design parameters, such as electrode size, were also investigated to determine best practices for hardware design of EQS-HBC systems. In these experiments, signal of a particular frequency is applied at the transmitter end and the received signal is measured through an oscilloscope. The transmitter device is on the wrist of the user and the received voltage is measured on the other hand. The channel loss is calculated from the ratio of the received (V_{Rx}) and transmitted voltage (V_{Tx}) (*Loss* = $20 \log_{10} \frac{V_{Rx}}{V_{Tx}}$). The selectivity and security properties are evaluated through signal leakage measurements at different positions away from the body. Since the signal leakage magnitude is very small, a spectrum analyzer, which provides more sensitivity



Fig. 6. Box and whisker plots of channel loss across different subjects with frequencies. The results show that the channel loss is almost independent of frequency and is primarily contained within -55 dB to -45 dB.

compared to oscilloscope, is used for measurements. The spectrum analyzer measures the power of the received signal (P_{Rx}). The noise floor (P_{noise}) of the spectrum analyzer determines the lowest signal that can be measured by the equipment. In a practical scenario, any signal under the noise floor cannot be recovered by a receiver. Hence, to establish the strength/quality of the received signal for leakage measurements we compare it with the noise floor and use Signal to Noise Ratio (SNR) as the measurement metric ($SNR = 10 \log_{10} \frac{P_{Rx}}{P_{noise}}$). Sections 4 and 5 use these measurement setups to evaluate the channel and leakage characteristics of the EQS-HBC channel. The results of this evaluation are used to determine the implementation details of the *BodyWire-HCI* system, particularly the receiver device. The implementation details are discussed later in Section 7.

4.1 Loss Variation Across Subjects

Channel loss measurements were acquired during different times of the day for seven users over an extended period of time. Environmental interferences and variation in ground coupling on the human body (posture) were strictly controlled in order to measure channel loss variation. External interferences present in the laboratory environment and picked up by the body were eliminated through passive filtering to accurately measure the received signal amplitude. The results are presented in Figure 6 to show the range of intra-body channel loss for seven different subjects (four male and three female) across multiple days in the frequency range of 50 kHz to 1 MHz.

The channel loss ranges between -55 and -45 dB for all users while implementing EQS-HBC. Using the data from this characterization, *BodyWire-HCI* is designed with an input range capable of handling this dynamic signal from all users and is robust against channel loss fluctuations between different users, as well as the fluctuations arising from use during different times of the day.

4.2 Loss Variation with Transmitter/Receiver Electrode Size

Electrode size is a critical factor in the design of an HBC device as the total device form factor of a wearable HBC device is critically dependent on the electrode size. During the design of *BodyWire*-*HCI*, the effect of electrode area on channel loss was studied for both the transmitter and receiver



Fig. 7. Channel loss variation with transmitter/receiver signal electrode size. The channel loss is almost independent of the electrode sizes with slightly extra loss for the smallest eletrode size $(0.5 \times 0.5 \text{ cm})$.

side. Channel loss measurements were carried out with different combinations of transmitter and receiver electrode size. The loss showed little variation depending on electrode size from $0.5 \text{ cm} \times 0.5 \text{ cm}$ up to $4 \times 4 \text{ cm}$. This is intuitive as the channel loss is dominated by the return path loss, dependent on the size of the floating ground electrode [48] and not strongly dependent on the variations in the forward path, of which the electrodes are a part of. The channel losses for all the combinations of electrode sizes are shown in the form of a bar graph in Figure 7.

5 DESIGN GOALS

5.1 Touch-Based Communication

One of the primary design goals of the BodyWire-HCI system is to show communication strictly during touch events. The transmitter and receiver should communicate only when there is a direct conductive path through the human body. This can only be achieved if there is sufficient difference in the amount of signal received between the cases of direct touch and leakage from an adjacent electrode. As a result, the receiver will be selectively sensitive to signals received directly through touch. For this, we have utilized signal transmission at frequencies of around 500 kHz. The low frequency operation enables EQS-HBC and results in minimal signal radiation out of the body. The measurements carried out in Figure 8 shows the viability of strictly touch-based communication. The experiments are carried out on two subjects (two male) and the results are the average of five different measurements over multiple days. In the experiments, electrodes of size 1×1 cm size are kept 3 mm away from each other. The user touches one of the electrodes and the received signal is measured through a spectrum analyzer at all of the electrodes. Figure 8 shows the variation in SNR of the received signal with frequency at all of the electrodes. It can be seen that the signal leakage to adjacent electrodes increases with signal transmission frequencies. The received signal at the touched electrode increases with frequency in this case due to the low input impedance (50 Ω) of the spectrum analyzer. However, at all the measured frequencies between 500 kHz and 20 MHz, we see a significant difference (>20 dB) in actual received signal and the leakage from adjacent electrodes. As a result, the receiver can be designed to be sensitive to received signal only when it is touched directly but insensitive to leakage from adjacent electrodes. This enables the selectivity



Fig. 8. Measurement of received SNR for different electrode configurations: User touching (a) electrode 1, (b) electrode 2, and (c) electrode 3, respectively, for a three electrode configuration. (d) User touching electrode 3 in a four electrode configuration. All measurements show more than 20dB degradation in SNR on adjacent electrodes compared to the electrode being touched enabling selectivity. The signal leakage to adjacent electrodes also significantly reduces at 500 kHz, compared to 20MHz, when measured using a spectrum analyzer with 50Ω input impedance.

of the touch events achievable through EQS-HBC in our current system design. Also it can be seen that the difference in SNR on the electrode being touched and the leakage to adjacent electrode gets larger as the frequency of operation reduces.

Furthermore, a characterization of hand position during transmission was conducted to ensure that *BodyWire-HCI* remains strictly touch based independent of the posture of the hand touching the electrode. A total of 24 participants were presented with an array of electrodes as well as the EQS-HBC system and asked to touch electrode 5 in the electrode array, as shown in Figure 9. The way in which the user's hand naturally selected the electrode could be characterized into three main hand postures as shown in Figure 9. Each hand posture elicits a distinct leakage pattern on the surrounding electrodes that are 3mm apart. Leakage measurements were conducted for each of the hand postures—leading to the SNR data shown in Figure 9. In addition, the most popular posture (posture 1) was tested with the user approaching the electrode from five different angles. Again, there was a distinct difference in leakage pattern as displayed in Figure 10. Although leakage varied due to hand position and angle, the SNR difference between the intended receive signal to the leakage never fell below 20 dB giving the ability for *BodyWire-HCI* to formulate heavily localized electrode arrays for data transmission or authentication.



Fig. 9. The effect of posture of the touching hand on signal leakage to adjacent electrodes. Measurement results show that the signal leakage is almost independent of the hand posture. The SNR at the electrode being touched (electrode 5) is also significantly larger than the SNR at the adjacent electrodes.



Fig. 10. (a) Characterization of leakage in surrounding electrodes as the user approaches the electrode array from the various angles with hand posture as in Posture 1. The SNR of the touched electrode and the leakage to the adjacent electrodes as the angle of approach is (b) 180° , (c) 225° , (d) 270° , (e) 315° , and (f) 360° .

ACM Transactions on Computer-Human Interaction, Vol. 27, No. 6, Article 39. Publication date: November 2020.



Fig. 11. The effect of transmitting arm posture on received signal strength. (a) The transmitting arm gradually moves closer to the torso from its outstretched position and (b) the received signal strength, hence the SNR, reduces as the arm moves closer to the torso.

5.2 Posture Independence

For signal transmission between a wrist worn/handheld device and a computer, the arm posture affects the amount of signal received at the receiver end. Specifically, the distance of the arm from the torso of the user affects the amplitude of the received voltage. Experiments in Figure 11 show four different postures of the arm with the transmitter and the corresponding received signal at the receiving electrode when the other hand is touching it. It can be seen that there is 7 dB difference in the amount of received signal depending on the posture of the arm. The maximum amount of signal is received for the posture with an outstretched arm. As the arm with the transmitter is brought closer to the torso, the received signal at the receiver end. The *BodyWire-HCI* system is required to be designed to operate across all these posture variations. The receiver is required to have sufficient sensitivity to operate during the posture with minimum received voltage but should not get saturated when the amount of received signal increases to maximum for a different posture.

5.3 Convenience

One of the primary goals of HBC is to enhance convenience during user interaction by combining communication with touch. The combination of communication and touch combines multiple user interaction steps into one single step, which enhances the ease of use. The *BodyWire-HCI* system is designed to provide secure communication during touch and enhance user convenience during HCI.

5.4 Wide Applicability

The *BodyWire-HCI* system serves as the communication backbone infrastructure for any application requiring transmission of information strictly through touch. The system is designed to take data from a transmitter device, transmit it through EQS-HBC, and decode it at the receiver end. The type of transmitter and receiver device does not affect the hardware/software of the



Fig. 12. Interferences picked up by the body during transmission in EQS-HBC. (a) 60 Hz mains power supply noise. (b) Additional 40 KHz interference present in the laboratory environment.

BodyWire-HCI system. This will allow wide applicability of the *BodyWire-HCI* system for different application scenarios such as data transfer and secure authentication.

5.5 Interference Tolerance

The human body picks up any interference present in the environment due to the human body antenna effect [11, 47]. This interference affects any ongoing HBC data transmission, resulting in bit errors. Since the frequency of operation of the *BodyWire-HCI* system is in the range of 100s of KHz, any environmental interference in this frequency range affects data transmission. Figure 12 shows two of the prominent source of interference present in the laboratory environment, as measured through an oscilloscope connected to the body. The 60 Hz noise (Figure 12(a)) originates from the supply mains within the building, an interference that will be present in an actual application scenario within any building. The interference around 40 KHz (Figure 12(b)) is due to the light sources present in the laboratory. The hardware of the *BodyWire-HCI* system is designed to filter out the different interferences present within the environment. The physical layer also needs to have error detection/correction mechanisms to tolerate bit errors happening due to any interference affecting transmission.

5.6 Form Factor

The *BodyWire-HCI* system is designed for applications involving HCI. Hence, one of the devices involved in the interaction will be a wearable or a handheld device. Therefore, both the transmitter and the receiver should be designed to have a wearable form factor. This puts a limit on the battery life and the processing power of the computing unit (microcontrollers) present on these devices.

6 DISCUSSION: SAFETY AND LIMITATION

6.1 Safety

HBC requires transmission of electrical signal through the human body. Hence safety limits of electrical signal transmission are one of the primary concerns of any HBC system. Every HBC-based system should be designed to adhere to the safety limits of current and radiation (Figure 13). The International Commission on Non-Ionizing Radiation Protection (ICNIRP) provides guidelines for limits of human exposure to time varying electric, magnetic and electro-magnetic fields up to 300 GHz frequency. Since the *BodyWire-HCI* system operates in the frequency range of a few 100s of KHz, we are primarily focused on the low frequency range. The RMS current density through the body is the primary bottleneck at this frequency range. The 4Hz–1kHz frequency range has the most stringent current density requirement as the current at these frequencies has effect on nerve



Fig. 13. (a) Safety limit table of current density and SAR provided by the ICNIRP guidelines. (b) Current density of EQS-HBC from simulated circuit models, showing >10000x lower current density compared to the safety limits.

stimulation. Figure 13(b) shows the safety requirement of current density for different frequencies. Simulating an EQS-HBC circuit model [33] shows that the current density in EQS-HBC is more than 10000x lower than the safety limits imposed by the guidelines [45]. This is primarily due to the high impedance provided by the return path capacitance at the low frequency of operation, limiting the overall current through the system. A detailed study about the safety of HBC can be found in [45].

6.2 Limitations

The *BodyWire-HCI* system enables strictly touch–based communication through the underlying hardware design. However, there is no way for the system to differentiate between the touch of an intended receiver and an unintended malicious attacker. An attacker can touch the user and receive the transmitted signal and there is no mechanism in place to stop that from happening since the transmitter does not depend on any feedback from the receiver during transmission. If we look into previous HBC implementations or other technologies such as Bluetooth, it is possible for an attacker to be in close proximity and still snoop the signal. Hence the possibility to attack from close physical proximity can be thwarted through *BodyWire-HCI*, significantly reducing the chance of an attacks. To summarize, although *BodyWire-HCI* does not protect the signal from touch-based attacks it nullifies any chance of nearby snooping.

7 SYSTEM LEVEL IMPLEMENTATION

The evaluation of the channel and leakage characteristics in Sections 4 and 5 is used to determine the design requirements of the *BodyWire-HCI* system, such as receiver sensitivity and maximum operating data rate for reliable operation. The complete *BodyWire-HCI* transceiver is implemented using Commercial Off-The-Shelf components, as shown in Figure 14(a). The receiver consists of an Analog Front End (AFE) designed on a custom Printed Circuit Board (PCB) for signal processing and a Digital Processing Unit (DPU) to run the software and interface with the external world. The AFE primarily consists of filters and amplifiers to process the received signal. The transmitted signal gets attenuated and is affected by environmental interference at the receiver end. Hence, appropriate analog signal processing needs to be performed to recover the received bits from the noisy, attenuated signal, before it can be utilized by the DPU. The amplifiers are designed from off the shelf Operational Amplifier IC (LM 324 series). The filters are designed from passive



Fig. 14. (a) BodyWireHCI device components: PCB of the analog front-end, TM4C123G microcontroller acting as the Digital Processing Unit, Lithium Poly Battery used as power source, and 3D printed casing for packaging the device. (b) Block diagram of the transmitter and receiver showing the basic building blocks and how they are interconnected.

resistances and capacitances. The DPU is implemented on a Texas Instruments TM4C123G microcontroller, which has a 32 bit, 80 MHz, ARM Cortex M4 processor. The board consists of 32 KB SRAM and 56 KB flash memory. Two 12 bit, 2 MSPS ADCs are used to sample the signal coming out of the AFE and these act as an interface between the AFE and the DPU. The 32 bit Timer internal to the microcontroller is used to provide precise timing necessary for ADC sampling operation. The DPU performs basic error correction on the received bits and decodes the data from its packetized form. The DPU also acts as an interface to the touchscreen display unit through the Serial Peripheral Interface. A BOOSTXL-K350QVG-S1 QVGA touchscreen display is used for detecting external touch input and displaying images or text. The complete block diagram of the receiver board is shown in Figure 14(b).

The transmitter also has a TM4C123G microcontroller unit as its DPU. The transmitter packetizes the data payload and transmits it through one of the General Purpose Input Output pin present in the microcontroller. The transmitter can also receive external input through touch screen display (BOOSTXL-K350QVG-S1) or push buttons present on the microcontroller board. Both the transmitter and receiver are powered through rechargeable Li-ion batteries to reduce power supply noise and comply with the wearable form factor. The data rate of transmission achievable in the prototype *BodyWire-HCI* system is 8 Kbps. The transmitter block diagram is also shown in Figure 14(b).

BodyWire-HCI uses a half-duplex channel through the body for communication. The transmitter and receiver use Time Division Multiplexing to transmit/receive data at alternate time intervals. It is possible to use Frequency Division Multiplexing to use the body as a duplex channel but has not been implemented in the current version. The synchronization between transmitter, receiver, and error detection is achieved through packet-based protocol. There are the following three different types of packets: (a) Preamble Packet: calibrates the receiver for decoding; (b) Header Packet: determines transmission size; and (c) Data Packet: Contains the transmitted data (64–1024 bit data in one packet). Error Correction is implemented to recover transmission errors. The maximum transmission length without calibration is limited by the clock frequency drift between the transmitter and receiver device.

8 APPLICATIONS

We show three example demonstrations of the *BodyWire-HCI* system and illustrate how it can be used for strictly touch–based communication. The first application demonstrates the selectivity and the strict touch requirement of communication through HBC, where the data intended for a



Fig. 15. Selective information transmission from a transmitter to a receiver: (a) Two receivers kept in close proximity to each other; and (b) Data transmitted to the receiver, whose electrode is touched by the user. The adjacent receiver does not receive the signal. Demo: https://www.youtube.com/watch?v=Uwrig2XQIH8.

particular receiver is not picked up by an adjacent receiver in close proximity. The second application shows transfer of information from a wrist-watch to a computer through the touch of an electrode, highlighting the convenience and entertainment potential. Lastly, we demonstrate the envisioned application of secure authentication through touch by unlocking a computer when a person wearing the appropriate key touches an electrode connected to the computer—highlighting an application where security is enhanced.

8.1 Selectivity: Data Transmission with Multiple Receivers

Selectivity is one of the key advantages of signal confinement within the body, achieved by *BodyWire-HCI*. The selectivity aspect of *BodyWire-HCI* is the focus of this current demonstration. In this scenario there are two receivers, which are kept in close proximity to each other. The transmitter transmits a code corresponding to the color chosen by the user on the touchscreen display. The data only gets transmitted to the receiver whose electrode is touched by the user (Figure 15). Although the electrode of the other receiver is in close proximity, it does not receive the signal (Figure 15(b)), showing selectivity provided by the *BodyWire-HCI* prototype. The signal confinement property of EQS-HBC makes selective information transmission possible, as shown through this demonstration. This selectivity property can be utilized for applications requiring precise location tracking and localization.

8.2 Data Transfer: Wearable to Computer Data Transfer

In this demonstration, we utilize the *BodyWire-HCI* framework to transfer information from a wearable device such as a smart-watch to a computer (Figure 16). A *BodyWire-HCI* receiver is connected to the computer, which uses serial USB to communicate with the receiving computer. When the user touches the electrode connected with the receiver, it forms a communication channel between the devices. The code corresponding to the image being displayed in the wearable gets transferred to the *BodyWire-HCI* receiver through EQS-HBC, which is subsequently transferred to the receiving computer through USB serial communication. The receiving computer displays the image corresponding to the code it receives on its display. This demonstration shows information exchange between a wearable transmitter and a computer. This can be extended to image transmission also but that requires higher data rates than 8kbps supported through the *BodyWire-HCI* prototype. The receiver device in this demonstration is implemented as a separate device connected to the computer through USB for the ease of implementation. However, it can always be integrated within the computer as a separate module with only the electrode acting as the external interface.



Fig. 16. Information transmission from a wearable device to a computer through touch. (a) User about to touch the electrode and (b) information transfer during touch of electrode resulting in image displayed in the computer.

This example illustrates the possible usage of *BodyWire-HCI* framework for entertainment purposes. Transferring data from one device to another just through touch provides a very natural and intuitive way of information exchange and enhances user convenience significantly. This demonstration can also be utilized for other usage scenarios such as downloading map information from computer to a smart-watch or using a body worn wearable as an intermediate device for data transfer between two computers.

8.3 Secure Authentication: Unlocking Computers

This demonstration of the *BodyWire-HCI* system shows a mock up demonstration of unlocking of a computer screen with the touch of an electrode connected to a HBC receiver, as seen in Figure 17(a)–(d). The computer unlocks only when the user with the correct key touches the electrode connected to the computer, through the HBC receiver. To demonstrate the unlocking process, the image transitions from the lock screen to the unlock screen only when the correct code is received from at the receiver connected to the computer. This is analogous to unlocking a computer only when a correct code is received from a wearable device, acting as an additional factor in a multi factor authentication scenario. It can also be seen from our experiments that the computer does not unlock when the fingertip of the user is very close (<1cm) to the receiver electrode but not touching it. This shows that there is very little signal leakage out of the body, demonstrating the additional physical layer security provided by EQS-HBC compared to radio communication. Hence, utilizing EQS-HBC is advantageous for applications like secure authentication as shown here.

This demonstration can also be seen as part of a two factor authentication system, where the unique key of the transmitter acts as a substitute to password. This can also be used for two factor authentication systems where the user has to use both his/her biometric identification (like fingerprint) and a passcode for authentication. This requires two separate steps of scanning fingerprint and entering the passcode. However, by doing communication through touch it is possible to perform two-factor authentication through fingerprint matching and also authenticating through the key sent by the transmitter. This can potentially enhance the convenience of such applications by reducing the number of steps performed by the user.

Through EQS-HBC, it is possible to utilize the human body as a physical communication channel, where the signal is primarily confined within the channel itself and there is minimal radiation out of the body. This enables additional physical layer security which can be utilized to create secure channels during pairing of devices, such as a streaming device and a TV. Traditionally, in



Fig. 17. Demonstration showing secure authentication of a computer through the key transmitted by a body worn device. (a) The computer is locked and does not get unlocked even when the user with correct key is in close proximity to the electrode. (b) Zoomed in diagram showing air gap between the finger and electrode. (c) Computer gets unlocked when the user with the correct key is touching the electrode. (d) Computer does not unlock when the user with the incorrect key touches the electrode.

wireless protocols such as Bluetooth, such pairing process begins by key exchange over non-secure air medium, making them vulnerable to malicious attacks. This demonstration shows the possibility of key exchange through EQS-HBC, which can be utilized for establishing secure channels for wireless communication.

9 CONCLUSION

Touch acts as a natural way of interaction between humans and computing resources present in the environment. Coupling communication with touch will enhance its effect significantly opening up new interaction modalities in HCIs. In this article, we demonstrate BodyWire-HCI, which uses EQS-HBC to minimize signal leakage out of the body and hence enable communication strictly during touch, previously not achievable through wireless radio wave communication or other HBC-based systems. The low frequency operation of EQS-HBC helps it reduce signal leakage out of the body significantly, enabling communication strictly through touch. Design techniques, such as capacitive termination and voltage mode transmission/reception have been used to minimize loss through the human body at low frequencies to enable EQS-HBC. The system has been designed to tolerate variation of signal strength among different users as well as for a user due to posture variation, time of day, and so on. It is also designed to be tolerant to interferences present at low frequencies. The *BodyWire-HCI* system has been used to show data transmission selectively and strictly during touch events and demonstrate applications such as transfer of an image from a wearable to a computer. BodyWire-HCI also shows promise for applications such as secure authentication of a computer through a body worn key. As computers increasingly become ubiquitous, systems coupling communication and touch will provide new modalities to interact with them. BodyWire-HCI provides such a platform for communication strictly during touch. We hope this

39:22

will stimulate future research on new HCI modalities which benefits from the coupling of communication and touch.

REFERENCES

- Shitij Avlani, Mayukh Nath, Shovan Maity, and Shreyas Sen. 2020. A 100KHz-1GHz termination-dependent human body communication channel measurement using miniaturized wearable devices. In *Proceedings of the 2020 Design, Automation Test in Europe Conference Exhibition (DATE'20)*. 650–653. DOI: https://doi.org/10.23919/DATE48585.2020. 9116556
- [2] J. Bae, H. Cho, K. Song, H. Lee, and H. Yoo. 2012. The signal transmission mechanism on the surface of human body for body channel communication. *IEEE Trans. Microw. Theory Tech.* 60, 3 (2012), 582–593. DOI: https://doi.org/10.1109/ TMTT.2011.2178857
- [3] J. Bae, K. Song, H. Lee, H. Cho, and H. Yoo. 2012. A 0.24-nJ/b wireless body-area-network transceiver with scalable double-FSK modulation. *IEEE J. Solid-State Circuits* 47, 1 (2012), 310–322. DOI: https://doi.org/10.1109/JSSC.2011. 2170632
- [4] Rafael Ballagas, Michael Rohs, and Jennifer G. Sheridan. 2005. Sweep and point and shoot: Phonecam-based interactions for large public displays. In Proceedings of the CHI 2005 Extended Abstracts on Human Factors in Computing Systems (CHI EA'05). ACM, New York, NY, 1200–1203. DOI: https://doi.org/10.1145/1056808.1056876
- [5] Doug A. Bowman, Ernst Kruijff, Joseph J. LaViola, and Ivan Poupyrev. 2001. An Introduction to 3-D user interface design. Presence Teleoper Virtual Env. 10, 1 (2001), 96–108. DOI: https://doi.org/10.1162/105474601750182342
- [6] Leah Buechley, David Mellis, Hannah Perner-Wilson, Emily Lovell, and Bonifaz Kaufmann. 2010. Living wall: Programmable wallpaper for interactive spaces. In *Proceedings of the 18th ACM International Conference on Multimedia* (*MM*'10). ACM, New York, NY, 1401–1402. DOI: https://doi.org/10.1145/1873951.1874226
- [7] M. A. Callejón, D. Naranjo-Hernández, J. Reina-Tosina, and L. M. Roa. 2013. A comprehensive study into intrabody communication measurements. *IEEE Trans. Instrum. Meas* 62, 9 (2013), 2446–2455. DOI: https://doi.org/10.1109/TIM. 2013.2258766
- [8] Niamh Caprani, Noel E. O'Connor, and Cathal Gurrin. 2012. Touch screens for the older user. In Assistive Technologies, Fernando A. Auat Cheein (Ed.). IntechOpen. DOI: https://doi.org/10.5772/38302. Available at https://www.intechopen. com/books/assistive-technologies/touch-screens-for-the-older-user.
- [9] Keith Cheverst, Alan Dix, Daniel Fitton, Chris Kray, Mark Rouncefield, Corina Sas, George Saslis-Lagoudakis, and Jennifer G. Sheridan. 2005. Exploring Bluetooth based mobile phone interaction with the hermes photo display. In Proceedings of the 7th International Conference on Human Computer Interaction with Mobile Devices & Services (MobileHCI'05). ACM, New York, NY, 47–54. DOI: https://doi.org/10.1145/1085777.1085786
- [10] H. Cho, H. Kim, M. Kim, J. Jang, J. Bae, and H. Yoo. 2015. 21.1 A 79pJ/b 80Mb/s full-duplex transceiver and a 42.5µW 100kb/s super-regenerative transceiver for body channel communication. *IEEE J. Solid-State Circuits* 51, 1 (2015), 310–317. DOI:https://doi.org/10.1109/ISSCC.2015.7063085
- [11] N. Cho, L. Yan, J. Bae, and H.J. Yoo. 2009. A 60 kb/s #x2013;10 Mb/s adaptive frequency hopping transceiver for interference-resilient body channel communication. *IEEE J. Solid-State Circuits* 44, 3 (2009), 708–717. DOI: https://doi. org/10.1109/JSSC.2008.2012328
- [12] Gabe Cohn, Sidhant Gupta, Tien-Jui Lee, Dan Morris, Joshua Smith, Matthew Reynolds, Desney Tan, and Shwetak Patel. 2012. An ultra-low-power human body motion sensor using static electric field sensing. In *Proceedings of the International Conference on Ubiquitous Computing (UbiComp'12)*. ACM, 99–102. DOI: https://doi.org/10.1145/2370216. 2370233
- [13] Gabe Cohn, Daniel Morris, Shwetak N. Patel, and Desney S. Tan. 2011. Your noise is my command: Sensing gestures using the body as an antenna. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems* (CHI'11). Association for Computing Machinery, 791–800. DOI: https://doi.org/10.1145/1978942.1979058
- [14] Gabe Cohn, Daniel Morris, Shwetak Patel, and Desney Tan. 2012. Humantenna: using the body as an antenna for real-time whole-body interaction. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems* (CHI'12). Association for Computing Machinery, 1901–1910. DOI :https://doi.org/10.1145/2207676.2208330
- [15] Debayan Das, Shovan Maity, Baibhab Chatterjee, and Shreyas Sen. 2019. Enabling covert body area network using electro-quasistatic human body communication. *Sci. Rep.* 9, 1 (2019), 4160. DOI:https://doi.org/10.1038/s41598-018-38303-x
- [16] Paul Dietz and Darren Leigh. 2001. DiamondTouch: A multi-user touch technology. In Proceedings of the 14th Annual ACM Symposium on User Interface Software and Technology (UIST'01). ACM, New York, NY, 219–226. DOI: https://doi.org/10.1145/502348.502389
- [17] Masaaki Fukumoto and Mitsuru Shinagawa. 2005. CarpetLAN: A novel indoor wireless(-like) networking and positioning system. In Proceedings of the 7th International Conference on Ubiquitous Computing (UbiComp'05). Springer, Berlin, 1–18. DOI: https://doi.org/10.1007/11551201_1

ACM Transactions on Computer-Human Interaction, Vol. 27, No. 6, Article 39. Publication date: November 2020.

BodyWire-HCI: Enabling New Interaction Modalities

- [18] Nan-Wei Gong, Steve Hodges, and Joseph A. Paradiso. 2011. Leveraging conductive inkjet technology to build a scalable and versatile surface for ubiquitous sensing. In *Proceedings of the 13th International Conference on Ubiquitous Computing (UbiComp'11)*. Association for Computing Machinery, 45–54. DOI:https://doi.org/10.1145/2030112. 2030120
- [19] Tobias Grosse-Puppendahl, Andreas Braun, Felix Kamieth, and Arjan Kuijper. 2013. Swiss-cheese extended: An object recognition method for ubiquitous interfaces based on capacitive proximity sensing. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI'13). ACM, New York, NY, 1401–1410. DOI: https://doi.org/ 10.1145/2470654.2466186
- [20] Tobias Grosse-Puppendahl, Christian Holz, Gabe Cohn, Raphael Wimmer, Oskar Bechtold, Steve Hodges, Matthew S. Reynolds, and Joshua R. Smith. 2017. Finding common ground: A survey of capacitive sensing in human-computer interaction. In *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems (CHI'17)*. Association for Computing Machinery, 3293–3315. DOI: https://doi.org/10.1145/3025453.3025808
- [21] Iman Habib, Niklas Berggren, Erik Rehn, Gustav Josefsson, Andreas Kunz, and Morten Fjeld. 2009. DGTS: Integrated typing and pointing. In Proceedings of the 12th IFIP TC 13 International Conference on Human-Computer Interaction: Part II (INTERACT'09). Springer, Berlin, 232–235. DOI: https://doi.org/10.1007/978-3-642-03658-3_30
- [22] Chris Hand. 1997. A survey of 3d interaction techniques. Comput. Graph. Forum 16, 5 (1997), 269-281. DOI: https://doi.org/10.1111/1467-8659.00194
- [23] Robert Hardy and Enrico Rukzio. 2008. Touch & interact: Touch-based interaction of mobile phones with displays. In Proceedings of the 10th International Conference on Human Computer Interaction with Mobile Devices and Services (MobileHCI'08). ACM, New York, NY, 245–254. DOI: https://doi.org/10.1145/1409240.1409267
- [24] Chris Harrison, Munehiko Sato, and Ivan Poupyrev. 2012. Capacitive fingerprinting: Exploring user differentiation by sensing electrical properties of the human body. In *Proceedings of the 25th Annual ACM Symposium on User Interface Software and Technology (UIST'12)*. ACM, New York, NY, 537–544. DOI: https://doi.org/10.1145/2380116.2380183
- [25] Mehrdad Hessar, Vikram Iyer, and Shyamnath Gollakota. 2016. Enabling on-body transmissions with commodity devices. In Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing (Ubi-Comp'16). ACM, New York, NY, 1100–1111. DOI: https://doi.org/10.1145/2971648.2971682
- [26] Christian Holz and Patrick Baudisch. 2013. Fiberio: A touchscreen that senses fingerprints. In Proceedings of the 26th Annual ACM Symposium on User Interface Software and Technology (UIST'13). ACM, New York, NY, 41–50. DOI: https: //doi.org/10.1145/2501988.2502021
- [27] Christian Holz and Frank R. Bentley. 2016. On-Demand Biometrics: Fast cross-device authentication. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems (CHI'16). ACM, New York, NY, 3761–3766. DOI: https://doi.org/10.1145/2858036.2858139
- [28] Christian Holz, Senaka Buthpitiya, and Marius Knaust. 2015. Bodyprint: Biometric user identification on mobile devices using the capacitive touchscreen to scan body parts. In *Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems (CHI'15)*. ACM, New York, NY, 3011–3014. DOI: https://doi.org/10.1145/2702123. 2702518
- [29] Christian Holz and Marius Knaust. 2015. Biometric touch sensing: Seamlessly augmenting each touch with continuous authentication. In Proceedings of the 28th Annual ACM Symposium on User Interface Software & Technology (UIST'15). ACM, New York, NY, 303–312. DOI:https://doi.org/10.1145/2807442.2807458
- [30] ikinamo. Human Body as a Communication Medium : DigInfo. Retrieved from https://www.youtube.com/watch?v= bgl88nrd7do.
- [31] B. Kibret, M. Seyedi, D. T. H. Lai, and M. Faulkner. 2014. Investigation of galvanic-coupled intrabody communication using the human body circuit model. *IEEE J. Biomed. Health Inform* 18, 4 (2014), 1196–1206. DOI:https://doi.org/10. 1109/JBHI.2014.2301165
- [32] Yuhwan Kim, Seungchul Lee, Inseok Hwang, Hyunho Ro, Youngki Lee, Miri Moon, and Junehwa Song. 2014. High5: Promoting interpersonal hand-to-hand touch for vibrant workplace with electrodermal sensor watches. In Proceedings of the 2014 ACM International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp'14). ACM, New York, NY, 15–19. DOI: https://doi.org/10.1145/2632048.2632072
- [33] Ubicomp Lab. Carpacio: Repurposing capacitive sensors to distinguish driver and passenger touches on in-vehicle screens. Retrieved February 5, 2019 from https://ubicomplab.cs.washington.edu/publications/carpacio/.
- [34] Guohao Lan, Dong Ma, Weitao Xu, Mahbub Hassan, and Wen Hu. 2017. CapSense: Capacitor-based activity sensing for kinetic energy harvesting powered wearable devices. In Proceedings of the 14th EAI International Conference on Mobile and Ubiquitous Systems: Computing, Networking and Services (MobiQuitous'17). ACM, New York, NY, 106–115. DOI:https://doi.org/10.1145/3144457.3144459
- [35] Chia-Hsun Lee and Ted Selker. 2004. iSphere: A proximity-based 3d input device. In Proceedings of the ACM SIGGRAPH 2004 Posters (SIGGRAPH'04). ACM, New York, NY, 72–. DOI: https://doi.org/10.1145/1186415.1186499

- [36] Ž. Lučev, I. Krois, and M. Cifrek. 2011. A capacitive intrabody communication channel from 100 kHz to 100 MHz. In Proceedings of the 2011 IEEE International Instrumentation and Measurement Technology Conference. 1–4. DOI: https://doi.org/10.1109/IMTC.2011.5944311
- [37] S. Maity, B. Chatterjee, G. Chang, and S. Sen. 2018. A 6.3pJ/b 30Mbps #x2212;30dB SIR-tolerant broadband interference-robust human body communication transceiver using time domain signal-interference separation. In Proceedings of the 2018 IEEE Custom Integrated Circuits Conference (CICC'18). 1–4. DOI: https://doi.org/10.1109/CICC. 2018.8357033
- [38] Shovan Maity, Baibhab Chatterjee, Gregory Chang, and Shreyas Sen. 2019. BodyWire: A 6.3-pJ/b 30-Mb/s -30-dB SIR-tolerant broadband interference-robust human body communication transceiver using time domain interference rejection. *IEEE J. Solid-State Circuits* 54, 10 (2019), 2892–2906. DOI: https://doi.org/10.1109/JSSC.2019.2932852
- [39] Shovan Maity, Debayan Das, Baibhab Chatterjee, and Shreyas Sen. 2018. Characterization and classification of human body channel as a function of excitation and termination modalities. In Proceedings of the 2018 40th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC'18). 3754–3757. DOI: https://doi.org/10.1109/EMBC.2018.8513332
- [40] Shovan Maity, Debayan Das, Xinyi Jiang, and Shreyas Sen. 2017. Secure human-internet using dynamic human body communication. In Proceedings of the 2017 IEEE/ACM International Symposium on Low Power Electronics and Design (ISLPED'17). 1–6. DOI: https://doi.org/10.1109/ISLPED.2017.8009190
- [41] Shovan Maity, Debayan Das, and Shreyas Sen. 2017. Wearable health monitoring using capacitive voltage-mode human body communication. In Proceedings of the 2017 39th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC'17). 1–4. DOI: https://doi.org/10.1109/EMBC.2017.8036748
- [42] Shovan Maity, Mingxuan He, Mayukh Nath, Debayan Das, Baibhab Chatterjee, and Shreyas Sen. 2019. Bio-Physical modeling, characterization, and optimization of electro-quasistatic human body communication. *IEEE Transactions* on Biomedical Engineering 66, 6 (2019), 1791–1802. DOI:10.1109/TBME.2018.2879462
- [43] Shovan Maity, Nirmoy Modak, David Yang, Shitij Avlani, Mayukh Nath, Josef Danial, Debayan Das, Parikha Mehrotra, and Shreyas Sen. 2020. A 415 nW physically and mathematically secure electro-quasistatic HBC Node in 65nm CMOS for authentication and medical applications. In *Proceedings of the 2020 IEEE Custom Integrated Circuits Conference* (CICC'20). 1–4. DOI: https://doi.org/10.1109/CICC48029.2020.9075930
- [44] Shovan Maity, Kavian Mojabe, and Shreyas Sen. 2018. Characterization of Human Body Forward Path Loss and Variability Effects in Voltage-Mode HBC. *IEEE Microw. Wirel. Compon. Lett.* 28, 3 (March 2018), 266–268. DOI: https://doi.org/10.1109/LMWC.2018.2800529
- [45] Shovan Maity, Mayukh Nath, Gargi Bhattacharya, Baibhab Chatterjee, and Shreyas Sen. 2020. On the Safety of Human Body Communication. *IEEE Trans. Biomed. Eng.* (2020), 1–1. DOI: https://doi.org/10.1109/TBME.2020.2986464
- [46] N. Matsushita, S. Tajima, Y. Ayatsuka, and J. Rekimoto. 2000. Wearable key: Device for personalizing nearby environment. In Proceedings of the 4th International Symposium on Wearable Computers. 119–126. DOI: https://doi.org/10. 1109/ISWC.2000.888473
- [47] Mayukh Nath, Shovan Maity, Shitij Avlani, Scott Weigand, and Shreyas Sen. 2020. Inter-body coupling in electroquasistatic human body communication: Theory and analysis of security and interference properties. ArXiv200313329 Cs Eess (2020).
- [48] Mayukh Nath, Shovan Maity, and Shreyas Sen. 2020. Toward understanding the return path capacitance in capacitive human body communication. *IEEE Trans. Circuits Syst. II Express Briefs* 67, 10 (2020), 1879–1883. DOI: https://doi.org/ 10.1109/TCSII.2019.2953682
- [49] Nextel Italia S.r.I. Tecnologia NFC RedTacton (En). Retrieved from https://www.youtube.com/watch?v=uAzphwZ8VLY.
- [50] Aditya Shekhar Nittala, Anusha Withana, Narjes Pourjafarian, and Jürgen Steimle. 2018. Multi-touch skin: A thin and flexible multi-touch sensor for on-skin input. In *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems (CHI'18)*. ACM, New York, NY, 33:1–33:12. DOI: https://doi.org/10.1145/3173574.3173607
- [51] Duck Gun Park, Jin Kyung Kim, Jin Bong Sung, Jung Hwan Hwang, Chang Hee Hyung, and Sung Weon Kang. 2006. TAP: Touch-and-play. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI'06). ACM, New York, NY, 677–680. DOI: https://doi.org/10.1145/1124772.1124873
- [52] J. Park and P. P. Mercier. 2015. Magnetic human body communication. In Proceedings of the 2015 37th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC'15). 1841–1844. DOI: https://doi.org/10.1109/EMBC.2015.7318739
- [53] Jun Rekimoto. 2002. SmartSkin: An infrastructure for freehand manipulation on interactive surfaces. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI'02). ACM, New York, NY, 113–120. DOI: https://doi.org/10.1145/503376.503397
- [54] W. Saadeh, M. A. B. Altaf, H. Alsuradi, and J. Yoo. 2017. A 1.1-mW ground effect-resilient body-coupled communication transceiver with pseudo OFDM for head and body area network. *IEEE J. Solid-State Circuits* 52, 10 (2017), 2690–2702. DOI: https://doi.org/10.1109/JSSC.2017.2713522

BodyWire-HCI: Enabling New Interaction Modalities

- [55] Dominik Schmidt, Ming Ki Chong, and Hans Gellersen. 2010. HandsDown: Hand-contour-based user identification for interactive surfaces. In Proceedings of the 6th Nordic Conference on Human-Computer Interaction: Extending Boundaries (NordiCHI'10). ACM, New York, NY, 432–441. DOI: https://doi.org/10.1145/1868914.1868964
- [56] Shreyas Sen. 2016. SocialHBC: Social networking and secure authentication using interference-robust human body communication. In Proceedings of the 2016 International Symposium on Low Power Electronics and Design (ISLPED'16). ACM, New York, NY, 34–39. DOI: https://doi.org/10.1145/2934583.2934609
- [57] Shreyas Sen. 2019. Wearable health monitoring system and method using human body communication. Retrieved August 1, 2020 from https://patents.google.com/patent/US20190347393A1/en.
- [58] Shreyas Sen. 2019. Human body communication interference rejection system. Retrieved August 1, 2020 from https://patents.google.com/patent/US20190379414A1/en.
- [59] Shreyas Sen and Shovan Maity. 2020. System and method for signal interference rejection using human body communication. Retrieved August 1, 2020 from https://patents.google.com/patent/US20200028532A1/en.
- [60] J. R. Smith. 1996. Field mice: Extracting hand geometry from electric field measurements. *IBM Syst. J.* 35, 3–4 (1996), 587–608. DOI: https://doi.org/10.1147/sj.353.0587
- [61] Masato Takahashi, Charith Lasantha Fernando, Yuto Kumon, Shuhey Takeda, Hideaki Nii, Takuji Tokiwa, Maki Sugimoto, and Masahiko Inami. 2011. Earthlings attack!: A ball game using human body communication. In Proceedings of the 2nd Augmented Human International Conference (AH'11). ACM, New York, NY, 17 1–17:4. DOI: https://doi.org/10.1145/1959826.1959843
- [62] M. Valtonen, J. Maentausta, and J. Vanhala. 2009. TileTrack: Capacitive human tracking using floor tiles. In Proceedings of the 2009 IEEE International Conference on Pervasive Computing and Communications. 1–10. DOI: https://doi.org/10. 1109/PERCOM.2009.4912749
- [63] Virag Varga, Gergely Vakulya, Alanson Sample, and Thomas R. Gross. 2018. Enabling interactive infrastructure with body channel communication. Proc. ACM Interact Mob. Wearable Ubiquitous Technol. 1, 4 (2018), 169:1–169:29. DOI:https://doi.org/10.1145/3161180
- [64] Virag Varga, Marc Wyss, Gergely Vakulya, Alanson Sample, and Thomas R. Gross. 2018. Designing groundless body channel communication systems: Performance and implications. In *Proceedings of the 31st Annual ACM Symposium on User Interface Software and Technology (UIST'18)*. ACM, New York, NY, 683–695. DOI: https://doi.org/10.1145/3242587. 3242622
- [65] M. S. Wegmueller, M. Oberle, N. Felber, N. Kuster, and W. Fichtner. 2010. Signal transmission by galvanic coupling through the human body. *IEEE Trans. Instrum. Meas.* 59, 4 (2010), 963–969. DOI:https://doi.org/10.1109/TIM.2009. 2031449
- [66] Ruey-Sing Wei and Alberto Sangiovanni-Vincentelli. 1985. PLATYPUS: A PLA test pattern generation tool. In Proceedings of the 22Nd ACM/IEEE Design Automation Conference (DAC'85). IEEE Press, Piscataway, NJ, 197–203. Retrieved from http://dl.acm.org/citation.cfm?id=317825.317856.
- [67] Eileen Wood, Teena Willoughby, Alice Rushing, Lisa Bechtel, and Jessica Gilbert. 2005. Use of computer input devices by older adults. J. Appl. Gerontol 24, 5 (2005), 419–438. DOI: https://doi.org/10.1177/0733464805278378
- [68] Yang Zhang, Junhan Zhou, Gierad Laput, and Chris Harrison. 2016. SkinTrack: Using the body as an electrical waveguide for continuous finger tracking on the skin. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems (CHI'16). ACM, New York, NY, 1491–1503. DOI: https://doi.org/10.1145/2858036.2858082
- [69] Junhan Zhou, Yang Zhang, Gierad Laput, and Chris Harrison. 2016. AuraSense: Enabling expressive aroundsmartwatch interactions with electric field sensing. In *Proceedings of the 29th Annual Symposium on User Interface* Software and Technology (UIST'16). ACM, New York, NY, 81–86. DOI: https://doi.org/10.1145/2984511.2984568
- [70] T. G. Zimmerman. 1996. Personal area networks: Near-field intrabody communication. IBM Syst. J. 35, 3.4 (1996), 609–617. DOI:https://doi.org/10.1147/sj.353.0609
- [71] 2016. Panasonic's wild new technology transmits data by human touch. PCWorld. Retrieved September 20, 2018 from https://www.pcworld.com/article/3128025/wearables/panasonic-has-a-technology-for-transmitting-data-byhuman-touch.html.
- [72] Touché: Enhancing touch interaction on humans, screens, liquids, and everyday objects. *Disney Research*. Retrieved February 5, 2019 from https://www.disneyresearch.com/publication/touche-enhancing-touch-interactionon-humans-screens-liquids-and-everyday-objects/.
- BodyCom Technology | Microchip Technology. Retrieved September 20, 2018 from http://www.microchip.com/ design-centers/embedded-security/technology/bodycom-trade-technology.
- [74] Investigating Human Body Communications | Intel Newsroom. Retrieved September 20, 2018 from https://newsroom. intel.com/editorials/investigating-human-body-communications-wearables/.

Received February 2019; revised April 2020; accepted June 2020